Author's response to reviews

Title: Syndromic surveillance: STL for modeling, visualizing, and monitoring disease counts

Authors:

Ryan P Hafen (rhafen@purdue.edu)
David E Anderson (danders2@xula.edu)
William S Cleveland (wsc@purdue.edu)
Ross Maciejewski (rmacieje@purdue.edu)
David S Ebert (ebertd@purdue.edu)
Ahmad Abusalah (aabusala@purdue.edu)
Mohamed Yakout (myakout@cs.purdue.edu)
Mourad Ouazzani (mourad@cs.purdue.edu)
Shaun J Grannis (sgrannis@regenstrief.org)

Version: 8 Date: 14 April 2009

Author's response to reviews:

All requested revisions have been completed:

Major revisions (We require the author to make these changes)

Reference 2 - please provide full details for this reference.

Reference 17 - please provide full details for this reference.

Figure citations - all figures must be cited in numerical order in the text. Figure 7 has only been cited as placement text, please list it in the main text as well. Also, cite each figure individually rather than using figure ranges. For example See figures 1, 2 and 3 rather than See figures 1-3.

Table citation - please cite Table 1 within the main text rather than just as placement text.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change The Title of My Prestigious Publication: A Scientific Study to The title of my prestigious publication: a scientific study.

Duplicate authors' names - please remove the list of authors names at the top of the title page. They already appear in their correct position beneath the title.
Country - Please remember to include the country in the affiliation details.

Figure and table placement text - Please remove this.

Table title and legend - The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.